Author's response to reviews

Title: High prevalence of germline STK11 mutations in Hungarian Peutz-Jeghers Syndrome patients

Authors:

Janos Papp (janos.papp@oncol.hu)
Marietta E Kovacs (ria@oncol.hu)
Szilvia Solyom (szilvia.solyom@oulu.fi)
Miklos Kasler (m.kasler@oncol.hu)
Anne-Lise Borresen-Dale (a.l.borresen-dale@medisin.uio.no)
Edith Olah (e.olah@oncol.hu)

Version: 6 Date: 30 November 2010

Author's response to reviews: see over
Dear BMC Series Editorial Production Team,

In reply to your e-mail of today (30th of Nov., 2010) we made all the requested changes on the final revised version of our manuscript:

**Major revisions** were made earlier for the requests of the reviewers, and these were accepted. No additional changes were made in this final version.

**Additional files** are renumbered and now cited in numerical order throughout the text and without using ranges, as requested,

**Minor revisions** were not made in this version.

**Typography** is checked thoroughly.

Thank you again for accepting our manuscript for publication in BMC Medical Genetics.

Sincerely,

Edith Olah